Author’s response to reviews

Title: Effects of Panax notoginseng saponins on severe acute pancreatitis through the regulation of mTOR/Akt and caspase-3 signaling pathway by upregulating miR-181b expression in rats

Authors:
Ming-wei Liu (lmw2004210@163.com)
Rui Wei (758437603@qq.com)
Mei-xian Su (Kmsnx24@126.com)
Wei Zhang (2436730979@qq.com)
Hui Li (3420220794@qq.com)
Tian-wen Fang (13890801975@163.com)

Version: 1 Date: 18 Jun 2017

Author’s response to reviews:

BCAM-D-17-00608

Effects of panax notoginseng saponins on severe acute pancreatitis through mTOR/Akt and caspase-3 signaling pathway by upregulating miR-181b expression in rats

Ming-wei Liu; Rui Wei; Mei-xian Su; Wei Zhang; Hui Li; Tian-wen Fang

BMC Complementary and Alternative Medicine

Dear PhD Tom Rowles

Thank you for receiving your comments. According to your comments, the overlap exists between the text of our manuscript and a number of previous publications was revised. According to the Reference checking, references 10,15,28,30,39,43 and 54 were revised, reference 20 was added.

We hope that the revisions in the manuscript will be sufficient to make our manuscript suitable for publication in BMC Complementary and Alternative Medicine.
We shall look forward to hearing from you at your earliest convenience.

Yours sincerely,

Prof Ming-wei Liu